Author's response to reviews

Title: Breakpoint characterization of a novel large intragenic deletion of MUTYH detected in a MAP patient: Case report

Authors:

Giovana T Torrezan (gitorrezan@hotmail.com)
Felipe C C da Silva (felipecavalcanticarneiro@gmail.com)
Ana Cristina V Krepischi (ana.krepischi@gmail.com)
Érika M M Santos (erikammsantos@gmail.com)
Fábio de O Ferreira (fabio.ferreira@hcancer.org.br)
Benedito M Rossi (rossibm@gmail.com)
Dirce M Carraro (dirce.carraro@accamargo.org.br)

Version: 2 Date: 11 July 2011

Author's response to reviews:

As requested, the following paragraph was added to the manuscript at the Case Presentation section: Written informed consent was obtained from all patients and controls. This study was approved by the ethics committee of A. C. Camargo Hospital (approval number: 1169/08-B). The consent section was also updated to include all patients and controls.